Author's response to reviews

Title: the effects of pulmonary rehabilitation versus chest physical therapy on the levels of fibrinogen and albumin in patients with lung cancer undergoing lung resection: a randomized clinical trial

Authors:

Maria Tereza A P Morano (ttmorano@hotmail.com)
Rafael Mesquita (rafaelmesquita14@ymail.com)
Guilherme P F da Silva (guilhermepinheiro87@hotmail.com)
Amanda S Araujo (amandafisio@hotmail.com)
Juliana Maria S Pinto (jumsp@hotmail.com)
Antero G Neto (anterogn@terra.com.br)
Cynthia Maria S Viana (cinthyav561@gmail.com)
Manoel O de Morais Filho (odorico@ufc.br)
Eanes D B Pereira (eanes@fortalnet.com.br)

Version: 5 Date: 9 July 2014

Author's response to reviews:

TO Executive editor

BMC Pulmonary Medicine

Fortaleza, 9 July 2014

Ms. Ref. No.: 1277257857121293

Title: the effects of pulmonary rehabilitation versus chest physical therapy on the levels of fibrinogen and albumin in patients with lung cancer undergoing lung resection: a randomized clinical trial

Response to the concerns

The legend was included for figure 2

We copyedit the paper to improve the style of written English. We used the editing service of Edanz.

Point 1

A trial registry was listed. Line 51

Point 2
The word important was changed for common. Line 59

Point 3
Reference 5 was included. Line 65

Point 4
The word neoplasm was changed to primary tumor. Line 67

Point 5
The nature of altered coagulation was included. Line 68

Point 6
One of the outcomes of the studies was stated. Line 72

Point 7
To clarify methodology we stated that 3 groups were assigned to the PR group and 3 to the CPT group. Line 86

Point 8
The sentence: during preliminary testing was included. Line 116

Point 9
As enrollment is part of the methodology it was moved from the results to the methods. Line 91

Point 10
P was described as statistically significant. Line 159

Point 11
The first sentence of the second paragraph of discussion was removed. Line 168

Point 12
The sentence was included: in contrast to studies on inflammation in COPD patients. Line 171

Point 13
Line 175 to 180 presents a new English editing

Point 14
Reference 26 and 27 were deleted. Line 184

Point 15
The relevant characteristics of the 1284 subjects were included. Line 188

Point 16

The sentence in line 201 was deleted and reference 34

Point 17

The sentence in line 213 and 214 were deleted

Point 18

Reference 26 and 27 was deleted. Now reference 28 and 29 will be the number 26 and 27. Line 187

Point 19

The reference 30 and 31 will be the number 28 and 29. Line 188

Point 20

The reference 32 will be the number 30. Line 190

Point 21

The reference 33 will be the number 31. Line 197

Point 22

The reference 35 will be the number 32 and the reference 36 will be the number 33. Line 210